Author’s response to reviews

Title: Variations of circulating cardiac biomarkers during and after anthracycline-containing chemotherapy in breast cancer patients

Authors:

Pierre Freres (pierre.freres@gmail.com)
Nassim Bouznad (Nassim.Bouznad@med.uni-muenchen.de)
Laurence Servais (laurence.servais@student.ulg.ac.be)
Claire Josse (c.josse@ulg.ac.be)
Stéphane Wenric (s.wenric@ulg.ac.be)
Aurélie Poncin (aponcin@chu.ulg.ac.be)
Jérôme Thiry (thiry.jerome@gmail.com)
Marie Moonen (m.moonen@chu.ulg.ac.be)
Cécile Oury (cecile.oury@ulg.ac.be)
Lancellotti Patrizio (plancellotti@chu.ulg.ac.be)
Vincent Bours (vbours@ulg.ac.be)
Guy Jerusalem (g.jerusalem@chu.ulg.ac.be)

Version: 1 Date: 15 Nov 2016

Author’s response to reviews:

Editor's comments:

1 - Additional files 1 and 2 references have been included in the Results section (page 7, line 1 and page 8, line 16, respectively).

2 - The name of the committee that approved the study has been specified (page 4, line 7).
The different sections under the heading 'Declarations' have been integrated in the Manuscript.